\documentclass[]{article}
\headheight=8pt     \topmargin=0pt
\textheight=624pt   \textwidth=432pt
\oddsidemargin=18pt \evensidemargin=18pt

\usepackage{amssymb,amsmath,amsthm}
\usepackage{epsfig}
\usepackage{subfigure}
\usepackage{enumerate}
\usepackage{color}
\usepackage{graphicx}
\usepackage{enumerate}

\DeclareMathOperator{\Tr}{Tr}
\pagestyle{plain}

\theoremstyle{plain}

\begin{document}

\title{MTH 352/652\\
Quiz \#1}
\date{}
\maketitle

\begin{enumerate}
\item Find a function $u(t,x)$ that satisfies the PDE
\begin{equation*}
u_{x}=u
\end{equation*}
 subject to the boundary condition $u(t,1)=1$.

\end{enumerate}
\end{document}